Author’s response to reviews

Title: The various clinical spectra of juvenile xanthogranuloma: Imaging for two case reports and review of the literature

Authors:

Michaela Höck (michaela.hoeck@i-med.ac.at)
Bernhard Zelger (bernhard.zelger@tirol-kliniken.at)
Gisela Schweigmann (gisela.schweigmann@tirol-kliniken.at)
Barbara Brunner (barbara.brunner@tirol-kliniken.at)
Bettina Zelger (Bettina.zelger@i-med.ac.at)
Gabriele Kropshofer (gabriele.kropshofer@tirol-kliniken.at)
Ursula Kiechl-Kohlendorfer (Ursula.kohlendorfer@i-med.ac.at)

Version: 2 Date: 21 Mar 2019

Author’s response to reviews:

All assistant editor comments have been considered carefully.

1. Consent for publication is clearly stated.
2. Discussion and Conclusions is combined.
3. Supplementary material is removed.
4. A clean version of manuscript is uploaded.

We hope that the manuscript is now potentially acceptable for publication in BMC Pediatrics.